Author's response to reviews

Title: Practical considerations to guide development of access controls and decision support for genetic information in electronic medical records

Authors:

Diana C Darcy (Diana.Darcy@hhs.sccgov.org)
Eleanor T Lewis (Eleanor.Lewis@va.gov)
Kelly E Ormond (kormond@stanford.edu)
David J Clark (David.Clark5@va.gov)
Jodie A Trafton (jodie.trafton@va.gov)

Version: 3 Date: 18 October 2011

Author's response to reviews: see over
October 18, 2011

Dear editors,

We are delighted to submit the final version of our manuscript for publication in BMC Health Services Research. The reviewer did not request any changes. We have made the following changes requested by the editorial production team:

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Discussion, Summary

The abstract has been reformatted with the correct headings.

Structure: this manuscript does not have the correct structure. Please reformat it to use these headings: Abstract; Background; Discussion; Summary; Abbreviations (if any); Competing interests; Authors’ contributions; Authors’ information (optional); Acknowledgements (optional); References

The manuscript has been reformatted with the correct structure.

References: Please make sure references are cited in numerical order. We notice references 5 and 6 are not cited within the main text. Please make sure references continue to be cited in numerical order once these references are cited.

In the process of the prior revision, these two references were moved to the correct location on the reference list and renumbered there but not renumbered in the text. They have now been numbered correctly in the text and all reference numbers have been checked.

Minor revisions requested:

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

All highlighting and tracking has been removed.

Abbreviations: Please format these in to a paragraph. Please make sure all abbreviations are spelled out and occur in alphabetical order.

The abbreviations have been placed in alphabetical order, spelled out, and formatted into a paragraph as requested.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Minor revisions have been incorporated into the manuscript for purposes of clarity and grammatical correctness.

Please do not hesitate to contact me with further questions or concerns.
Sincerely,

Eleanor T. Lewis, Ph.D.
Center for Health Care Evaluation
VA Palo Alto Health Care System
795 Willow Rd. (152)
Menlo Park, CA 94025
Eleanor.Lewis@va.gov
(650) 493-5000 x 2-22949